Author's response to reviews

Title: Improving the management of non-ST elevation acute coronary syndromes: systematic evaluation of a quality improvement programme
European Quality Improvement Programme for Acute Coronary Syndrome: The EQUIP-ACS project

Authors:

Marcus D Flather (m.flather@rbht.nhs.uk)
Jean Booth (j.booth@rbht.nhs.uk)
Daphne S Babalis (d.babalis09@imperial.ac.uk)
Hector Bueno (hbueno.hugm@salud.madrid.org)
Phillipe G Steg (gabriel.steg@bch.aphp.fr)
Grzegorz Opolski (grzegorz.opolski@wum.edu.pl)
Filippo Ottani (ottani@alice.it)
Jacques Machecourt (jmachecourt@chu-grenoble.fr)
Alfredo Bardaji (abardaji@comt.es)
Mats Bojestig (Mats.Bojestig@lj.se)
Anthony R Brady (tony@sealedenvelope.com)
Bertil Lindahl (bertil.lindahl@ucr.uu.se)

Version: 5 Date: 13 January 2010

Author's response to reviews:

Revisions have been made to the Appendix section of the Manuscript as requested by the Editorial team at Trials Journal.